Author's response to reviews

Title: Effectiveness and cost-effectiveness of a group-based pain self-management intervention for patients undergoing total hip replacement: feasibility study for a randomised controlled trial

Authors:

Vikki Wylde (V.Wylde@bristol.ac.uk)
Elsa Marques (e.marques@bristol.ac.uk)
Neil Artz (neil.artz@bristol.ac.uk)
Ashley Blom (ashley.blom@bristol.ac.uk)
Rachael Gooberman-Hill (r.gooberman-hill@bristol.ac.uk)

Version: 3 Date: 28 April 2014

Author's response to reviews:

We have made the below requested minor formatting changes to the manuscript:

1. We have included the date the study was registered with the trial registration number at the bottom of the Abstract.

2. We have included a list of abbreviations used and their meanings, after the Conclusion.

3. We have moved the funding statement to the Acknowledgements section.

4. We have submitted a populated CONSORT checklist and flow diagram as additional files.